Author's response to reviews

Title: A Randomised Controlled Trial of Total Hip Arthroplasty Versus Resurfacing Arthroplasty in the Treatment of Young Patients with Arthritis of the Hip Joint

Authors:

Juul Achten (j.achten@warwick.ac.uk)
Nick R Parsons (nick.parsons@warwick.ac.uk)
Richard P Edlin (R.P.Edlin@leeds.ac.uk)
Damian Griffin (Damian.griffin@warwick.ac.uk)
Matthew L Costa (Matthew.costa@warwick.ac.uk)

Version: 2 Date: 10 December 2009

Author's response to reviews:

The following lines have been included in the methods/design section of the manuscript:

This study has been reviewed by the Coventry Research Ethics Committee under reference number 07/Q2802/26. The study was approved on the 9th of May 2007. The research carried out is in compliance with the Helsinki Declaration.